
\bibitem{gsw1} S.L. Glashow, \npb{22}{1961}{579}; \\
              S. Weinberg, \prl{19}{1967}{1264};\\
              A. Salam, in \qt{Elementary Particle Theory}, Stockholm 1968.
\bibitem{kobma1} M. Kobayashi and T. Maskawa, \ptp{49}{1973}{652}.
\bibitem{cdf1} F. Abe et cal. [CDF Coll.], \prl{74}{1995}{2626}; \\
S. Abachi et al. [D0 Coll.], \prl{74}{1995}{2632}. 
\bibitem{lefr} J. Lefran\c{c}ois, Proceedings Intern. Europhysics Conference
 on High Energy Physics, Marseille 1993.
\bibitem{eacco} E. Accomando et al., DESY--ECFA study: ``Physics with
              $e^+e^-$ Linear Colliders'',  DESY 97--100, {\it Phys. Rep.}
{\bf C} in press. 
\bibitem{PIK}P. Igo-Kemenes, J.H. K\"uhn [conv.], in DESY 93-123 A,B.
\bibitem{jarls1} C. Jarlskog (ed.), \qt{CP Violation}, World Scientific;
              see in particular the contribution by
              I.I.Y. Bigi, V.A. Khoze,
              N.G. Uraltsev and A.I. Sanda in this compendium.
\bibitem{bethk1} S. Bethke, A. Ricker and P.M. Zerwas,
                    \zpc{49}{1991}{59}.
\bibitem{fritz1} H. Fritzsch, Workshop on Strong Coupling Gauge
              Theories and Beyond, Nagoya, 1990, Report MPI--PAE/PTh
              64/90, and references quoted therein.
\bibitem{buchm1} W. Buchm\"uller in Ref.\cite{kuehn1}.
\bibitem{kuehn1} J.H. K\"uhn, P.M. Zerwas [conv.] et al., \qt{Z physics at LEP
                 1}, CERN 89--08.
\bibitem{reya1}  E. Reya, P.M. Zerwas [conv.] et al., in Proceedings ECFA
              LHC Workshop, Aachen 1990, CERN 90--10.
\bibitem{Zersaar} P.M. Zerwas, Top Physics at \epem Linear
                 Colliders, Proceedings Saariselk\"a Workshop;
                 DESY 91-108; \\
                 J.H. K\"uhn, Proceedings  2nd Intern. Workshop on
              Physics and Experiments with Linear Colliders, Waikoloa
               1993.

\bibitem{SLACschool}
J.H. K\"uhn, in
Proceedings SLAC Summer Institute on Particle Physics, Stanford  1995,
       SLAC-R-494 CONF-9507258-UC-414.
 
\bibitem{donog1} J.F Donoghue, D. Treille [conv.] et al., Proceedings of the
              Workshop \qt{High--Luminosity Extension of LEP }, CERN 1990.
\bibitem{ckkpr}  K.G. Chetyrkin, A. Kwiatkowski, and J.H. K\"uhn,
              \pr{277}{1996}{189}. 
\bibitem{CKT} K.G. Chetyrkin, A.L. Kataev and F.V. Tkachov,
            \plb{85}{1979}277;\\
         M. Dine and J. Sapirstein, \prl{43}{1979}{668};\\
         W. Celmaster and R.J. Consalves, \prl{44}{1980}560.
\bibitem{Kat} S.G. Gorishny, A.L. Kataev and  S.A. Larin,
            \plb{259}{1991}{144}.\\
L.R. Surguladze and M.A. Samuel, \prl{66}{1991}560.\\
 K.G. Chetyrkin, \plb{391}{1997}{402}.
\bibitem{knieh1} B.A. Kniehl and J.H. K\"uhn, \plb{224}{1989}{229};\\
                 B.A. Kniehl and J.H. K\"uhn, \npb{329}{1990}{547}.
\bibitem{ChK308}
 K.G. Chetyrkin and  J.H. K\"uhn, \plb{308}{1993}{127}.
\bibitem{CT327}
 K.G. Chetyrkin and  O.V. Tarasov, \plb{327}{1994}{114};
\\
S.A. Larin, T. van Ritbergen and J.A.M. Vermaseren, \plb{320}{1994}{159}.

\bibitem{schwi1} J. Schwinger, ``Particles, Sources and Fields''
                 Vol.II; 
                   Addison--Wesley 1973;\\
 G. K\"all\'en and A. Sabry,
                 K. Dan. Vidensk. Selsk. Mat.-Fys. Medd. {\bf 29}
                 (1955) No. 17.
\bibitem{jersa1} J. Jersak, E. Laermann and P.M. Zerwas,
               \plb{98}{1981}{363},
                 and \prd{25}{1982}{1218}.
\bibitem{chety1} K.G. Chetyrkin and J.H. K\"uhn, \plb{248}{1990}{359};\\
                 K.G. Chetyrkin, A. Kwiatkowski and J.H. K\"uhn,
\plb{282}{1992}{221};\\
 K.G. Chetyrkin and A. Kwiatkowski, \plb{319}{1993}{307}. 
\bibitem{hoa452}
A.H. Hoang,  J.H. K\"uhn, and T. Teubner, \npb{452}{1995}{173}.

\bibitem{cks371}
 K.G. Chetyrkin,  J.H. K\"uhn,  and M. Steinhauser,
\plb{371}{1996}{93},
\npb{482}{1996}{213}, and hep-ph/9705254.
\bibitem{beta3} 
T. van Ritbergen, J.A.M. Vermaseren, and S.A. Larin, hep-ph/9701390.
\bibitem{gamma3}
K.G. Chetyrkin, hep-ph/9703278;\\
T. van Ritbergen, J.A.M. Vermaseren, and S.A. Larin, hep-ph/9703284.

\bibitem{Schilcher} N. Gray, D.J. Broadhurst, W. Grafe and K. Schilcher,
                  \zpc{48}{1990}{673}.

\bibitem{HJKT}
A.H. Hoang, M. Je\.zabek, J.H. K\"uhn, and T. Teubner,
\plb{325}{1994}{495}; 
\\
M.H. Seymour, \npb{436}{1995}{163}.
\bibitem{OPAL353}
Opal Coll., R. Akers et al., \plb{353}{1995}{595}.
\bibitem{DELPHI97}
Delphi Coll., P. Abreu et al., CERN/PPE-97-039.

\bibitem{akhun1} A. Akhundov, D. Bardin and T. Riemann,
               \npb{276}{1988}{1};\\
              J. Bernabeu, T. Pich and A. Santamaria,
              \plb{200}{1988}{200};\\
              W. Beenakker and W. Hollik, \zpc{40}{1988}{141};\\
              B. Lynn and R. Stuart, \plb{252}{1990}{676}.
\bibitem{djoua1} A. Djouadi, J.H. K\"uhn and P.M. Zerwas,
               \zpc{46}{1990}{411}; \\
            A.~Djouadi, B.~Lampe and P.M.~Zerwas, \zpc{67}{1995}{123}.
\bibitem{arbuz} A.B.~Arbuzov, D.~Yu.~Bardin and A.~Leike, 
{\it Mod. Phys Lett.} {\bf 7} (1992) 2029 and Erratum {\bf 9} (1994) 1515.
\bibitem{mvelt} M. Veltman, \npb{123}{1977}{89}.
\bibitem{seidel} S. Eidelmann and F. Jegerlehner, \zpc{67}{1995}{585}.
\bibitem{DVER}
A. Djouadi and C. Verzegnassi, \plb{195}{1987}{265}.
\bibitem{KKST}
B.A. Kniehl, J.H. K\"uhn, and R.G. Stuart, \plb{214}{1988}{621}.
\bibitem{AFMT}
L. Avdeev, J. Fleischer, S. Mikhailov, and O.V. Tarasov,
               \plb{336}{1994}{560}, Erratum \plb{349}{1995}{597}.

\bibitem{CKSt}
 K.G. Chetyrkin,  J.H. K\"uhn,  and M. Steinhauser,
\plb{351}{1995}{331}, \prl{75}{1995}{3397}.
\bibitem{FTJR}
J. Fleischer, O.V. Tarasov, F. Jegerlehner, and P. Raczka,
 \plb{293}{1992}{437}; \\
A. Kwiatkowski and M. Steinhauser, \plb{344}{1995}{359}.
\bibitem{czk}
A. Czarnecki and J.H. K\"uhn, \prl {77}{1996}{3955};\\
A. Czarnecki and K. Melnikov, hep-ph/9609302, TTP96-33, 
  to appear in  {\it Phys. Rev.} {\bf D}. 

\bibitem{rclare}  LEP Electroweak Working Group, R. Clare et al., ``A
Combination of Preliminary Electroweak Measurements and Constraints on
the Standard Model'', CERN Report LEPEWWG/97--01. 
\bibitem{girar1} G. Girardi, W. Hollik and C. Verzegnassi,
              \plb{240}{1990}{492}.
\bibitem{denne1} A. Denner, J. Guth, W. Hollik and J. K\"uhn,
                     \zpc{51}{1991}{695}.
\bibitem{Leike} A.A. Akhundov,
              D.Yu. Bardin and A. Leike, \plb{261}{1991}{321};\\
             A.B. Arbuzov, D.Yu. Bardin and A. Leike, PHE-91-012.
\bibitem{Rieman}D.Yu. Bardin et al., Dubna Preprint E2-88-324.
\bibitem{JKSW} J.H. K\"uhn and R.G. Stuart, \plb{200}{1988}{360}; \\
    S. Jadach, J.H. K\"uhn, R.G. Stuart and Z. Was, \zpc{38}{1988}{609}.
\bibitem{JW} Z. W\c as and S. Jadach, \plb{219}{1989}{103}.
\bibitem{Citoun} J.E. Campagne and R. Zitoun, \zpc{43}{1989}{469}.
\bibitem{Burgers} F.A. Berends, G.J.H. Burgers and W.L. van Nerven,
                 \npb{297}{1988}{429}; Erratum \npb{304}{1988}{92}.
\bibitem{bigi1}  I. Bigi, \plb{155}{1985}{125}.
\bibitem{Danilov} M.V. Danilov, Proceedings of the LP-HEP 91 Conference,
                  Geneva 1991.
\bibitem{Abe}{K. Abe, Proceedings, Intern. Conference on HEP, Singapore 1990.}
\bibitem{kuehn2} J.H. K\"uhn, A. Reiter and P.M. Zerwas,
                   \npb{272}{1986}{560}.
\bibitem{Sehkozer} G. K\"opp, L.M. Sehgal
                   and P.M. Zerwas, \npb{123}{1977}{77}.
\bibitem{Bjork} J.D. Bjorken, \prd{40}{1989}{1513}.
\bibitem{czarn} A. Czarnecki, M. Jezabek, T. K\"orner and T.H. K\"uhn,
                 \prl{73}{1994}{384}; \\ 
G. Bonvicini and L. Randall, \prl{73}{1994}{392}.
\bibitem{Dalitz} R.H. Dalitz, G.R. Goldstein and R. Marshall,
                    \zpc{43}{1989}{441}.
\bibitem{field1} R.D. Field and R.P. Feynman, \npb{136}{1978}{1}.
\bibitem{ali1}   A. Ali, E. Pietarinen, G. Kramer and J. Willrodt,
            \plb{93}{1980}{155}.
\bibitem{ander1} B. Andersson, G. Gustafson,
              G. Ingelman and T. Sj\"ostrand, \prc{97}{1983}{33}.
\bibitem{march1} G. Marchesini and B.R. Webber, \npb{310}{1988}{461}.
\bibitem{altar1} G. Altarelli and G. Parisi, \npb{126}{1977}{298}.
\bibitem{bigi2}  I. Bigi, Y.L. Dokshitzer, V.A. Khoze,
J. K\"uhn and P. Zerwas, \plb{181}{1986}{157}.
\bibitem{mele1}  B. Mele and P. Nason, \plb{245}{1990}{635}.
\bibitem{Roude} P.~Roudeau, Proceedings, Lepton--Photon Symposium, Geneva 1991.
\bibitem{bjork1} J.D. Bjorken, \prd{17}{1978}{171}.
\bibitem{peter1} C. Peterson, D. Schlatter, I. Schmitt and P. Zerwas,
                  \prd{27}{1983}{105}.
\bibitem{artru1} X. Artru and G. Mennessier, \npb{70}{1974}{93}.
\bibitem{morri1} D.A. Morris, \npb{313}{1989}{634}.
\bibitem{ander2} B. Andersson, G. Gustafson and B. S\"oderberg,
                \zpc{20}{1983}{317}.
\bibitem{bowle1} M.G. Bowler, \zpc{11}{1981}{169}.
\bibitem{campb1} N.A. Campbell, A. Huntley and C. Michael,
                \npb{306}{1988}{51};\\
              K.D. Born, E. Laermann, R. Sommer, T.F.
              Walsh and P.M. Zerwas, \plb{329}{1994}{325}.
\bibitem{koern1} J.G. K\"orner, G. Schierholz and J. Willrodt,
               \npb{185}{1981}{365}.
\bibitem{bengt1} M. Bengtsson and P.M. Zerwas, \plb{208}{1988}{306}.
\bibitem{bara} R.~Barate et al. [Aleph Coll.], CERN--PPE/97--002 and {\it Z. Phys.} C in press.
\bibitem{adeva2} B. Adeva et al. [L3 Coll.], \plb{248}{1990}{227}.
\bibitem{loenn1} L. L\"onnblad, C. Peterson
            and T. R\"ognvaldsson, \prl{65}{1990}{1321};
          \npb{349}{91}{675}.
\bibitem{nacht1} O. Nachtmann and A. Reiter, \zpc{16}{1982}{45}.
\bibitem{barte1} W. Bartel et al., \plb{101}{1981}{129};
           and \plb{134}{1984}{275};\\
           H. Aihara et al., \zpc{28}{1985}{31}.
\bibitem{azimo1} Ya.I. Azimov, Yu.L. Dokshitzer, V.A. Khoze
       and S.I. Troyan, \plb{165}{1985}{147}.


\bibitem{keu}
V. Barger, K. Cheung, and W.Y. Keung, \prd{41}{1990}{1541}.

\bibitem{cit1}
G.T. Bodwin, E. Braaten, and G.P. Lepage, \prd{51}{1995}{1125}.

\bibitem{keu2}
K. Cheung, W.Y. Keung, and T.C. Yuan, \prl{76}{1996}{877};
\\
P. Cho, \plb{368}{1996}{171}.

\bibitem{branco} G.C. Branco, H.P. Nilles
               and K.H. Streng, \plb{85}{1979}{269}.


\bibitem{Stirling}
K. Hagiwara et al., \plb{267}{1991}{527}; Erratum \plb{316}{1993}{631}.

\bibitem{OPAL}
Opal Collaboration, G. Alexander et al., \plb{384}{1996}{343}.

\bibitem{Driesen}
V.M. Driesen, J.H. K\"uhn, and E. Mirkes, \prd{49}{1994}{3197}, 
and references therein.

\bibitem{bra}
E. Braaten and Yu-Qi Chen, \prl{76}{1996}{730}.

\bibitem{Fritsch}
H. Fritzsch and J.H. K\"uhn, \plb{90}{1980}{164}.


\bibitem{Amiri} F. Amiri and C.R. Ji, \prd{35}{1987}{3318}.
\bibitem{Bra} E. Braaten, K. Cheung, and T.C. Yuan,
\prd{48}{1993}{R5049}. 
\bibitem{Kis} V.V. Kiselev, A.K. Likhoed, M.V. Shevlyagin,
Phys. Atom. Nucl. {\bf 57} (1994) 689.

\bibitem{Leike2}
A. Leike and R. R\"uckl, Nucl. Phys.  (Proc. Supp.) {\bf 37B} (1994)
215, and references therein.
 
\bibitem{DelNT} Delphi Coll., P. Abreu et al., CERN-PPE/96-194.

\bibitem{Shepherd} 
C.H. Shepherd-Themistocleous, Proceedings, 28th Intern. Conference on HEP, Warsaw
1996, Z. Ajduk and A. Wr\'oblewski (eds.), World Scientific. 
\bibitem{Ratoff} 
P.N. Ratoff, ibid., p. 884.
\bibitem{Richman} 
J.D. Richman, ibid., p. 143.

\bibitem{bosma1} M. Bosman and R. R\"uckl in Ref.~\cite{kuehn1}
            and references therein.
\bibitem{NeubertStech} M. Neubert and B. Stech,
      hep-ph/9705292,  in {\em Heavy Flavours II}, World Scientific
            (1997),  A.J. Buras and M. Lindner (eds.).

\bibitem{BurasFleischer} A. Buras and J. Fleischer, hep-ph/9704376, in
{\em Heavy Flavours II}, World Scientific (1997),  A.J. Buras and
M. Lindner (eds.).
\bibitem{Datta} B. Datta, E.A. Paschos
                 and U. T\"urke, \plb{196}{1987}{382}.

\bibitem{Gibbons}
L.K. Gibbons,  Proceedings, 28th Intern. Conference on HEP, Warsaw
1996, \\ Z. Ajduk and A. Wr\'oblewski (eds.), World Scientific.


\bibitem{gratt1} G. Gratta et al., \nca{103}{1990}{821}.
%References  "Evidence and Mass bounds"

\bibitem{albre1} H. Albrecht et al. (ARGUS), \plb{192}{1987}{245}; \\
            M. Artuso et al. (CLEO), \prl{62}{1989}{2233}.
\bibitem{cquigg} C. Quigg, ``Top--ology'', Report FERMILAB--Pub--97/091--T.
\bibitem{cern96-183}  LEP Electroweak Working Group, R. Clare et al., ``A Combination of Preliminary Electroweak Measurements and Constraints on the Standard Model'', CERN Report CERN--PPE/96--183.
\bibitem{kaue1}  G.L. Kane and M.E. Peskin, \npb{195}{1982}{29}.
\bibitem{albaj3} C. Albajar et al. [UA1 Coll.], \plb{255}{1991} and Erratum \plb{262}{1991}{497}.
\bibitem{SchailZ} D.~Schaile and P.M.~Zerwas, \prd{45}{1992}{3262}.
\bibitem{veltm1} M. Veltman, \app{8}{1977}{475}.
\bibitem{chano1} M. Chanowitz, M. Furman and I. Hinchliffe,
                 \npb{153}{1979}{402}.
\bibitem{cabib1} N. Cabibbo, L. Maiani, G. Parisi and R. Petronzio,
                 \npb{158}{1979}{296}.
\bibitem{lindn1} M. Lindner, \zpc{31}{1986}{295}.
%*** Bis hierher  Referenzen fuer "Evidence and  Mass bounds".
\bibitem{acta} J.H.~K\"uhn, \app{12}{1981}{347};\\
               J.H.~K\"uhn, \apa { Suppl.XXIV}{1982}{203}.
\bibitem{stre} J.H.~K\"uhn and K.H.~Streng, \npb{198}{1982}{71}.
\bibitem{jk2}  M.~Je\.zabek and J.H.~K\"uhn, \npb{320}{1989}{20}.
\bibitem{czar2}
     A.~Czarnecki, M.~Je\.zabek and J.H.~K\"uhn,
\npb{351}{1991}{70}; \\
     A.~Czarnecki, M.~Je\.zabek, J.G. K\"orner, and J.H.~K\"uhn,
\prl{73}{1994}{384}; \\
     A.~Czarnecki and M.~Je\.zabek, \npb{427}{1994}{3}.
 
\bibitem{jk239} M.~Je\.zabek and J.H.~K\"uhn, \plb{239}{1994}{317}.

\bibitem{schmitt} M. Schmitt, in Workshop Report on $e^+e^-$ Linear Colliders DESY 96-123D, \\ p. 67, P.M. Zerwas (ed.).

\bibitem{jk3}  M.~Je\.zabek and J.H.~K\"uhn, \npb{314}{1989}{1}.
\bibitem{den1} A.~Denner and T.~Sack, \npb{358}{1991}{46}.
\\
A.~Czarnecki, \plb{252}{1990}{467}.
\\
C.S.~Li, R.J.~Oakes, T.C.~Yuan, \prd{43}{1991}{3759}.

\bibitem{smith}
B.~H. Smith and M.~B. Voloshin, \plb{340}{1994}{176}.

\bibitem{ac95a}
A.~Czarnecki, \app{26}{1995}{845}.

\bibitem{BLM}
S.~J. Brodsky, G.~P. Lepage, and P.~B. Mackenzie, \prd{28}{1983}{228}.

\bibitem{czprl76}
A. Czarnecki, \prl{76}{1996}{4124}.
\bibitem{czmel}
A. Czarnecki and K. Melnikov, TTP97-08, hep-ph/9703277 and TTP97-05,
hep-ph/9703291. 
\bibitem{czar3}
     A.~Czarnecki, M.~Je\.zabek and J.H.~K\"uhn, \app{20}{1989}{961}.
\bibitem{eil}
  G.~Eilam,R.R.~Mendel, R.~Migneron and A.~Soni, \prl{66}{1991}{3105}.
\bibitem{DennerHoang} 
A. Denner and A.H. Hoang, Nucl. Phys. {\bf B397} (1993) 483.

\bibitem{JezK}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \prd{48}{1993}{R1910};
\newblock E: ibid. {\bf D49} (1994) 4970.

\bibitem{ns30} G.L.~Kane, Proceedings, Madison Workshop 1979; \\
          I.~Bigi et al.\ in Ref.\cite{bigi2}; \\
          J.F.~Gunion, H.E.~Haber, G.~Kane and S.~Dawson, {\it The Higgs
          Hunter's Guide}, Addison--Wesley 1990; \\
          C.S.~Li and T.C.~Yuan, \prl{42}{1990}{3088}.
\bibitem{ns38} J.~Gunion et al., Ref.\cite{ns30}; \\
          A.~Mendez and A.~Pomarol, \plb{252}{1990}{461}; \\
          C.S.~Li and R.J.~Oakes, \prd{43}{1991}{855}.
\bibitem{ns31} K.~Hikasa and M.~Kobayashi, \prd{36}{1987}{724}.
\bibitem{borz} F. M. Borzumati and N. Polonsky, in Workshop Report 
on $e^+e^-$ Linear Colliders DESY 93-123D, p. 41, P.M. Zerwas (ed.). 
\bibitem{ns40} J.~Ellis and S.~Rudaz, \plb{128}{1983}{248}; \\
          A.~Bouquet, J.~Kaplan and C.~Savoy, \npb{262}{1985}{299}.
\bibitem{Venturi} A. Venturi, Workshop Report on $e^+e^-$ Linear 
Colliders DESY 96-123D, p. 269, P.M. Zerwas (ed.).
\bibitem{ns41} G.~Eilam, J.L.~Hewett and A.~Soni, \prd{44}{1991}{1473}.
\bibitem{ns32} H.~Fritzsch, \plb{224}{1989}{423}.
\bibitem{ns33} W.~Buchm\"uller and M.~Gronau, \plb{220}{1989}{641}.
 
\bibitem{gluec1} M. Gl\"uck, J.F. Owens and E. Reya,
                \prd{15}{1978}{2324};\\
                 B.L. Combridge, \npb{151}{1979}{429}.
\bibitem{wille1} S. Willenbrock and D.A. Dicus, \prd{34}{1986}{155};\\
                 C.--P. Yuan, \prd{41}{1990}{42}.
\bibitem{nason1} P. Nason, S. Dawson and R. K. Ellis,
                   \npb{303}{1988}{607};
                 {\em ibid.} {\bf B}327 (1989) 49.
\bibitem{beena1} W. Beenakker, H. Kuijf, W. L. van Neerven and J. Smith,
                    \prd{40}{1989}{54}.
\bibitem{meng1} R. Meng, G. A. Schuler, J. Smith and W. L. van Neerven,
              \npb{339}{1990}{325}.
\bibitem{guesk1} S. G\"usken, J. H. K\"uhn and P. M. Zerwas,
                    \plb{155}{1985}{185}.
\bibitem{fadin1} V. Fadin, V. Khoze, \jl{46}{1987}{417};
                   \sovnp{48}{1988}{487}; \\
                   V. Fadin, V. Khoze and T. Sj\"ostrand, CERN-TH 5687/90.
\bibitem{laen} E. Laenen, J. Smith and W.L. van Neerven, \npb{369}{1992}{543}.
\bibitem{berg} E.L. Berger and H. Contopanagos, \plb{361}{1995}{115}.
\bibitem{cata} S. Catani, M. Maugani, P. Nason and L. Trentadue, \plb{378}{1996}{329}.
\bibitem{holli2}  W. Beenakker, A. Denner, W. Hollik and T. Sack,
                 \npb{411}{1994}{343}.
\bibitem{elli} R.K. Ellis, \plb{259}{1991}{492}.
\bibitem{foley1} K.J. Foley et al., Workshop on Experiments,
                 Detectors and Experimental Areas for the Supercollider,
                 Berkeley 1987.
\bibitem{fayar1} L. Fayard and G. Unal in Ref.\cite{reya1}.
\bibitem{fusi} O.J.P.~Eboli, G.C.~Marques, S.F.~Novales and
                A.A.~Natale, \prd{34}{1986}{771}; \\
               S.~Dawson and S.~Willenbrock, \npb{284}{1987}{449}; \\
               C.P.~Yuan, \npb{310}{1988}{1}; \\
               R.P.~Kauffman, \prd{41}{1990}{3343}; \\
                H.~Veltman, \prd{43}{1991}{2236}.
\bibitem{hagi}
       K.~Hagiwara, H.~Murayama and T.~Watanabe, \npb{367}{1991}{257}.
\bibitem{djo} A.~Djouadi, J.~Kalinowski and P.M.~Zerwas, DESY 91-104.

\bibitem{Marti} M. Martinez and S. Orteu, in Workshop Report on $e^+e^-$ Linear
Colliders, DESY 96-123D, p. 75, P.M. Zerwas (ed.).
\bibitem{con} M.~Consoli, W.~Hollik and F.~Jegerlehner, in ``$Z$ Physics
              at LEP~1'', G.~Altarelli (ed.), CERN89-08.
\bibitem{exac}
          J.~Jersak, E.~Laermann and P.M.~Zerwas, \prd{25}{1982}{363};\\
J.~Schwinger, ``Particles, Sources and Fields'', Addison-Wesley (1973);\\
         L.~Reinders, H.~Rubinstein and S.~Yazaki, \prc{127}{1985}{1}.
\bibitem{appr} J.~Schwinger in \cite{exac};\\
         S.~G\"usken, J.H.~K\"uhn and P.~Zerwas, \plb{155}{1985}{185}.
\bibitem{CKS} 
K.G. Chetyrkin, J.H. K\"uhn, and M. Steinhauser, \npb{482}{1996}{213}.
\bibitem{CHKST}
K.G. Chetyrkin, A. Hoang, J.H. K\"uhn,  M. Steinhauser, and
                T. Teubner, \plb{384}{1996}{233}.
\bibitem{CKS2}
K.G. Chetyrkin, J.H. K\"uhn, and M. Steinhauser, TTP97-18,
                hep-ph/9705254. 
\bibitem{HKT}
A. Hoang, J.H. K\"uhn,  and
                T. Teubner, \npb{452}{1995}{173}.
\bibitem{been1} W.~Beenakker, S.C.~van~der~Marck
              and W.~Hollik, \npb{365}{1991}{24}.
\bibitem{JKsim}
M. Je\.zabek and J.H. K\"uhn, Phys. Lett. {\bf B316} (1993) 360.
\bibitem{GKKS}
B. Grzadkowski, P. Krawczyk, J.H. K\"uhn, and R.G. Stuart, Nucl. Phys. 
{\bf B281} (1987) 18.


\bibitem{fr2} G.~Jikia, \plb{257}{1991}{196}.
\bibitem{fr3} T.~Sj\"ostrand and P.M.~Zerwas, Proceedings, $e^+e^-$
       Collisions at 500 GeV: The Physics Potential, DESY 92--123A.
\bibitem{st1} G.L.~Kane, G.A.~Ladinsky and C.--P.~Yuan, \prd{45}{1992}{124}.
\bibitem{st2} C.~Schmidt and M.E.~Peskin,
             Workshop on Physics and Experiments with Linear Colliders,
            Saariselk\"a 1991.
\bibitem{st3} W.~Bernreuther et al. in Ref.\cite{PIK};\\
 W.~Bernreuther, A. Brandenburg and P. Overmann, DESY 96-123D, p. 49. 
\bibitem{st4} A.~Devoto, J.~Pumplin, W.~Repko and G.L.~Kane,
            \prl{43}{79}{1062}; 
\bibitem{st5} G.L.~Kane, Proceedings, SLAC Summer Institute 1990.
\bibitem{jk41} J.H. K\"uhn, \npb{237}{1984}{77}.

\bibitem{ginz} I.F.~Ginzburg et al., \nim{205}{1983}{47}; \\
        V.I.~Telnov in Proc. Workshop ``Physics with Linear Colliders'',
        Saariselk\"a  1991, and references therein.
\bibitem{rich} F.~Richard, LAL91-62 and references therein.
\bibitem{schr} S.Y.~Choi and F.~Schrempp, DESY 91-086.
\bibitem{steeg} J.~Steegborn, Diploma Thesis, University of Karlsruhe.



\bibitem{fk}
V.S.~Fadin and V.A.~Khoze, \jet{46}{1987}{525},


\bibitem{apple}
T.~Appelquist, M.~Dine and I.~Muzinich, \prd{17}{1978}{2074}.
\bibitem{sus}
L.~Susskind, in ``Weak and Electromagnetic Interactions at High
Energies'', Les Houches (1977). 
\bibitem{fis}
W.~Fischler, \npb{129}{1977}{157}.
\bibitem{bil}
A.~Billoire, \plb{92}{1980}{343}.

\bibitem{MPeter}
M. Peter, \prl{78}{1997}{602}. 

\bibitem{pesk}
J.M.~Strassler and M.E.~Peskin, \prd{43}{1991}{1500}.
\bibitem{JKT}  M.~Je\.zabek, J.H.~K\"uhn and T.~Teubner,
   \zpc{56}{1992}{653}.

\bibitem{TJ} M.~Je\.zabek and T.~Teubner, \zpc{59}{1993}{669}.
\bibitem{Buchm}
For a pedagogical discussion see e.g.~the 
appendix of W. Buchm\"uller and S.-H. H. Tye, Phys. Rev. {\bf D24} (1981) 132.

\bibitem{Fujii}
K.~Fujii, Lectures presented at SLAC Summer Institute on
Particle Physics, Stanford 1995.

\bibitem{Igo}
P. Igo-Kemenes, M. Martinez, R. Miguel, and S. Orteu,  in Proceedings,
{\em \epem\ Collisions 
at 500 GeV: The Physics Potential}, Part C;  P.M. Zerwas (ed.), 
DESY 93-123C, p. 319.
\bibitem{GuthK}
R. Guth and J.H. K\"uhn,  Nucl. Phys. {\bf B368} (1992) 38.

\bibitem{Comas}
P. Comas, R. Miquel, M. Martinez, and S. Orteu, DESY 96-123D, p. 59.
\bibitem{Sumino} Y.~Sumino, K.~Fujii, K.~Hagiwara, H.~Murayama, C.-K.~Ng,
   \prd{47}{1993}{56}

\bibitem{pedag} For a detailed and pedagogical discussion of the
interplay between the QCD potential and the momentum distribution see
M. Je\.zabek, J.H. K\"uhn, and T. Teubner, DESY 93--123C, p. 303, and
\cite{SLACschool}. 

\bibitem{hjkt}
    R. Harlander, M. Je\.zabek, J.H. K\"uhn and T. Teubner,
    \plb{346}{1995}{137}.

\bibitem{MurSum2}
    H. Murayama and Y. Sumino, \prd{47}{1993}{82}.

\bibitem{ttp48}
R. Harlander, M. Je\.zabek, J.H. K\"uhn, and M. Peter,  Z.~Phys. {\bf
C73} (1997) 477.
\bibitem{MelYak1} K. Melnikov and O. Yakovlev, \plb{324}{1994}{217}.
\bibitem{sumph} Y. Sumino, PhD Thesis, University of Tokyo, 1993
(unpublished). 
\bibitem{Huff} R.W.  Huff, {\it Ann. Phys.} {\bf 16} (1961) 288.
\bibitem{FMY94}
    K. Fujii, T. Matsui and Y. Sumino, \prd{50}{1994}{4341}.

\bibitem{Uber} H. \"Uberall, {\it Phys. Rev.} {\bf 119} (1960) 365. 
\bibitem{MKum} W. M\"odritsch and W. Kummer, \npb{430}{1994}{3}.













